Author's response to reviews

Title: Rapid chemokinetic movement and the invasive potential of lung cancer cells; a functional molecular study

Authors:

Sandra YY Fok (sfok5959@mail.usyd.edu.au)
Jeffrey S Rubin (rubinj@mail.nih.gov)
Fiona Pixley (pixley@aecom.yu.edu)
John Condeelis (condeeli@aecom.yu.edu)
Filip Braet (filip.braet@emu.usyd.edu.au)
Lilian L Soon (lilian.soon@emu.usyd.edu.au)

Version: 5 Date: 6 June 2006

Author's response to reviews: see over
Dear Editor,

Thank you again for your previous correspondence. We have addressed all the points raised and made the appropriate changes as follows:

**Item 1: Country - Please remember to include the country in the affiliation details.**

The country in the affiliation details has now been included. P1, L8 now reads

"1NANO, Australian Key Centre for Microscopy and Microanalysis, Electron Microscope Unit, University of Sydney, Sydney, NSW, 2006, Australia.

2Laboratory of Cellular and Molecular Biology, NCI, National Institutes of Health, Bethesda, MD 20892, USA.

3Department of Anatomy and Structural Biology, Albert Einstein College of Medicine of Yeshiva University, Bronx, NY 10461, USA."

**Item 2: Section headings: Please use sentence case for all headings and sub-headings in the manuscript (i.e. remove all unnecessary capitals).**

All unnecessary capitals have been removed from all headings and sub-headings.

**Item 3: Website addresses - Please cite website addresses in the reference list and remove them from the manuscript text. Please remove the web address from page 9.**

All cite website addresses have been removed from the manuscript text and placed in the reference list. For example: website reference from P 9 L 21 has now been moved to the reference list P22, L22. It reads


Any references in the text were also updated to reflect those changes.

**Item 4: Tables - Table 1 has been divided into a and b. It is important that tables are numbered in ascending numerical order - 1, 2 and 3. Can we please ask that the tables are renumbered and any references to the tables in the text are updated to reflect these changes.**

Table 1 has now be divided into 2 separate tables (Table 1 and Table 2) found on P28, therefore adding an extra table in the manuscript. All references to the tables in the text have also been updated reflecting the changes performed.
Item 5: **Figures** - *It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image.*

All images have now been cropped as closely as possible.

We look forward to hearing from you.

Best regards,

Lilian Soon